Author's response to reviews

Title: Spectral domain optical coherence tomography in patients after successful management of postoperative endophthalmitis following cataract surgery by pars plana vitrectomy

Authors:

Otto A Maneschg (ottomaneschg@gmail.com)
Éva Volek (volekeva@gmail.com)
János Németh (nemeth.janos@med.semmelweis-univ.hu)
Gábor M Somfai (somfaigm@yahoo.com)
Zsuzsanna Géhl (zsuzsannagehl@gmail.com)
Irén Szalai (szalaiiren@gmail.com)
Miklós D Resch (miklosresch@gmail.com)

Version: 2
Date: 14 March 2014

Author's response to reviews:

Dear Alice Murray

Executive editor

BMC Ophthalmology,

Please find enclosed our revised manuscript entitled “Spectral domain optical coherence tomography in patients after successful management of postoperative endophthalmitis following cataract surgery by pars plana vitrectomy”, which we would like to be considered for publication in your prestigious journal.

We have corrected our manuscript according to your instruction:

The following changes has been performed in the manuscript and uploaded:

a) Acknowledgements
b) line numbers and page numbers
c) reference citations
d) we put the Figure legends after the Table Section

We thank You very much for your help and we hope that our manuscript will get a positive evaluation.

Sincerely
Otto Alexander Maneschg MD

corresponding author